%File content.tex, 9 September, 1992
%\magnification=\magstep1
\input ij.sty
\input mssymb
\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=7pt %This controls the spacing between indexitems of
%both kinds
\tenpoint
\null\vskip 3cm

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}

\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\smc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
\def\indexauthor{}
\def\endindexitem{}
\def\indexpage{}
\def\indextitle{}

\def\indexsee\indexauthor#1\partner#2\endindexitem
{{\smc #1\unskip\enspace}{\sl See }{\smc #2\unskip}\par}

\index
\indexvolume 106
\indexyear 1998
\endindextitle

\indexitem 
\indexauthor Amberg, B. and L.~S.~Kazarin
\indextitle On finite products of soluble groups
\indexpage 93
\endindexitem

\indexitem
\indexauthor Cepedello Boiso, M.
\indextitle Approximation of Lipschitz functions by\break $\Delta$-convex
functions in Banach spaces
\indexpage 269
\endindexitem

\indexsee
\indexauthor De Le\'on, M.
\partner M.~Fern\'andez
\endindexitem

\indexitem
\indexauthor Dix, J.~G.
\indextitle Infinite frequency solutions to discontinuous control
systems with variable delay
\indexpage 165
\endindexitem

\indexitem
\indexauthor Fern\'andez, M., R.~Ib\'a\~nez and M.~de Le\'on
\indextitle The canonical spectral sequences for Poisson manifolds
\indexpage 133
\endindexitem

\indexitem
\indexauthor H\'ajek, P.
\indextitle Polynomial algebras on classical Banach spaces
\indexpage 209
\endindexitem

\indexsee
\indexauthor Ib\'a\~nez, R.
\partner M.~Fern\'andez
\endindexitem

\indexitem
\indexauthor Jim\'enez-Seral, P. and J.~P.~Lafuente
\indextitle On complemented non-\break abelian chief factors of a finite
group
\indexpage 177
\endindexitem 

\indexitem
\indexauthor Jonoska, N.
\indextitle A conjugacy invariant for reducible sofic shifts and its
semigroup characterizations
\indexpage 221
\endindexitem

\indexitem
\indexauthor Kamae, T.
\indextitle Linear expansions, strictly ergodic homogeneous cocycles
and fractals
\indexpage 313
\endindexitem

\indexsee  
\indexauthor Kazarin, L.~S.
\partner B.~Amberg
\endindexitem

\indexitem
\indexauthor Koldobsky, A.
\indextitle An application of the Fourier transform to sections of
star bodies
\indexpage 157
\endindexitem

\indexsee
\indexauthor Lafuente, J.~P. 
\partner P.~Jim\'enez-Seral
\endindexitem

\indexitem
\indexauthor Ledet, A.
\indextitle Embedding problems with cyclic kernel of order 4
\indexpage 109
\endindexitem

\indexitem
\indexauthor Leung, K.~H. and S.~H.~Man
\indextitle On Schur rings over cyclic groups
\indexpage 251
\endindexitem

\indexitem 
\indexauthor Linial, N., A.~Magen and M.~E.~Saks
\indextitle Low distortion Euclidean\break embeddings of trees
\indexpage 339
\endindexitem

\indexitem 
\indexauthor Magaard, K., K.~Strambach and H.~V\"olklein
\indextitle Finite quotients of the pure symplectic braid group
\indexpage 13
\endindexitem

\indexsee
\indexauthor Magen, A.
\partner N.~Linial
\endindexitem

\indexsee
\indexauthor Man, S.~H.
\partner K.~H.~Leung
\endindexitem

\indexitem
\indexauthor Morris, G. and T.~Ward
\indextitle Entropy bounds for endomorphisms\break commuting with $K$
actions
\indexpage 1
\endindexitem

\indexitem
\indexauthor Newelski, L.
\indextitle $\cal M$-gap conjecture and m-normal theories
\indexpage 285
\endindexitem

\indexsee
\indexauthor Saks, M.~E.
\partner N.~Linial
\endindexitem

\indexsee
\indexauthor Strambach, K.
\partner K.~Magaard
\endindexitem

\indexsee
\indexauthor V\"olklein, H.
\partner K.~Magaard
\endindexitem

\indexsee
\indexauthor Ward, T.
\partner G.~Morris
\endindexitem

\indexitem
\indexauthor Wasserman, A.
\indextitle A derivation HNN construction for Lie algebras
\indexpage 79
\endindexitem

\indexitem
\indexauthor Weiss, B.
\indextitle Finite dimensional representations and subgroup actions
on homogeneous spaces
\indexpage 189
\endindexitem

\indexitem
\indexauthor Zinoviev, D.
\indextitle Relation of orbital integrals on SO(5) and PGL(2)
\indexpage 29
\endindexitem

%End of file, content.tex
